%File content.tex, 9 September, 1992
%\magnification=\magstep1
\input ij.sty
\input mssymb
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=7pt %This controls the spacing between indexitems of
%both kinds
\tenpoint
\null\vskip 3cm

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}

\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\smc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
\def\indexauthor{}
\def\endindexitem{}
\def\indexpage{}
\def\indextitle{}

\def\indexsee\indexauthor#1\partner#2\endindexitem
{{\smc #1\unskip\enspace}{\sl See }{\smc #2\unskip}\par}

\def\dj{d\kern-.30em\raise0.65ex\vbox{\hrule width .3em height
.03em}}
\def\Dj{D\rlap{\kern-.70em\raise0.75ex
\vbox{\hrule width .3em height .03em}}}

\index
\indexvolume 98
\indexyear 1997
\endindextitle


\indexitem 
\indexauthor Adams, J.
\indextitle Character of the oscillator representation
\indexpage 229
\endindexitem

\indexitem
\indexauthor Bainov, D., Yu.~I.~Domshlak and P.~S.~Simeonov
\indextitle On the\break oscillation properties of first-order impulsive
differential equations with a deviating argument
\indexpage 167
\endindexitem

\indexsee
\indexauthor Bogdanovi\'c, S.
\partner M.~\'Ciri\'c
\endindexitem

\indexitem
\indexauthor Brodsky, S.~D. and J.~Howie
\indextitle The universal torsion-free image of a group
\indexpage 209
\endindexitem

\indexitem
\indexauthor Brozovic, D.~P. and R.~M.~Solomon
\indextitle On groups of hyperbolic\break length
\indexpage 61
\endindexitem

\indexitem
\indexauthor \'Ciri\'c, M. and S.~Bogdanovi\'c
\indextitle The lattice of positive quasi-orders  on a semigroup
\indexpage  157
\endindexitem 

\indexsee
\indexauthor De Sam Lazaro, J.
\partner M.~Lema\'nczyk
\endindexitem

\indexitem
\indexauthor deLaubenfels, R., Z.~Huang, S.~Wang and Y.~Wang
\indextitle Laplace\break transforms of polynomially bounded vector-valued
functions and\break semigroups of operators
\indexpage 189
\endindexitem

\indexsee  
\indexauthor Domshlak, Yu.~I.
\partner D.~Bainov
\endindexitem

\indexitem
\indexauthor \Dj ur\dj evi\'c, M.
\indextitle Generalized braided quantum groups
\indexpage 329
\endindexitem

\indexitem
\indexauthor Forrest, A.~H.
\indextitle $K$-groups associated with substitution minimal\break systems
\indexpage 101
\endindexitem

\indexsee
\indexauthor Howie, J.
\partner S.~D.~Brodsky
\endindexitem

\indexsee
\indexauthor Huang, Z.
\partner R.~deLaubenfels
\endindexitem

\indexitem 
\indexauthor Klemes, I. and K.~Reinhold
\indextitle Rank one transformations with\break singular spectral type
\indexpage 1
\endindexitem

\indexitem
\indexauthor Kudla, S.~S. and W.~J.~Sweet, Jr.
\indextitle Degenerate principal series\break representations for
$U(n,n)$
\indexpage 253
\endindexitem

\indexitem 
\indexauthor Lema\'nczyk, M. and J.~de Sam Lazaro
\indextitle Spectral analysis of certain compact factors for
Gaussian dynamical systems
\indexpage 307
\endindexitem

\indexitem
\indexauthor Liardet, P. and D.~Voln\'y
\indextitle Sums of continuous and differentiable functions in
dynamical systems
\indexpage 29
\endindexitem

\indexitem
\indexauthor Maharam, D.
\indextitle Spectral representation of generalized transition\break
kernels
\indexpage 15
\endindexitem

\indexitem
\indexauthor Mochizuki, K. and R.~Suzuki
\indextitle Critical exponent and critical blow-up for quasilinear
parabolic equations
\indexpage 141
\endindexitem

\indexsee
\indexauthor Reinhold, K.
\partner I.~Klemes
\endindexitem

\indexsee
\indexauthor Simeonov, P.~S.
\partner D.~Bainov
\endindexitem

\indexsee
\indexauthor Solomon, R.~M.
\partner D.~P.~Brozovic
\endindexitem

\indexsee
\indexauthor Suzuki, R.
\partner K.~Mochizuki
\endindexitem

\indexsee
\indexauthor Sweet, W.~J. Jr.
\partner S.~S.~Kudla
\endindexitem

\indexsee
\indexauthor Voln\'y, D.
\partner P.~Liardet
\endindexitem

\indexsee
\indexauthor Wang, S.
\partner R.~deLaubenfels
\endindexitem

\indexsee
\indexauthor Wang, Y.
\partner R.~deLaubenfels
\endindexitem

\endindex
%End of file, content.tex
